Author's response to reviews

Title: Influence of the incremental step size in work rate on exercise response and gas exchange in patients with pulmonary hypertension: a prospective controlled study

Authors:

Sven Glaeser (glaeser.sven@gmail.com)
Sven Lodziewski (sven.lodziewski@uni-greifswald.de)
Beate Koch (kochb@uni-greifswald.de)
Christian F. Opitz (c-opitz@web.de)
Henry Volzke (kochb@uni-greifswald.de)
Ralf Ewert (ewert@uni-greifswald.de)

Version: 2 Date: 19 September 2007

Author's response to reviews:

Thank you very much for your reply to the manuscript. The study we did was not planed as an interventional study. We prospectively investigated on exercise tests as a single center experience. the aim was to describe the comparison of two different exercise regimens. So far we did not register the study according to your mentioned criteria prior to the study.

We still hope the study is still suitable for your journal.